Author's response to reviews

Title: Excretion of urinary histamine and N-tele methylhistamine in patients with gastrointestinal food allergy compared to non-allergic controls during an unrestricted diet and a hypoallergenic diet

Authors:

Martin Raithel (martin.raithel@uk-erlangen.de)
Alexander Hagel (alexander.hagel@uk-erlangen.de)
Heinz Albrecht (heinz.albrecht@uk-erlangen.de)
Yurdaguel Zopf (yurdaguel.zopf@uk-erlangen.de)
Andreas Naegel (andreas.naegel@uk-erlangen.de)
Hanns-Wolf Baenkler (hanns-wolf.baenkler@uk-erlangen.de)
Fred Buchwald (fred.buchwald@synlab.com)
Hans-Wolfgang Schultis (wolfgang.schultis@synlab.com)
Juergen Kressel (juergen.kressel@uk-erlangen.de)
Eckhart G Hahn (eckhart.hahn@uk-erlangen.de)
Peter Konturek (innere_slf@thueringen-kliniken.de)

Version: 7
Date: 24 February 2015

Author's response to reviews:

Point-to-point description of the changes made:
1. Full names and email addresses of all authors included (page 1)
2. Abstract and keywords formatted according to the journal’s IforA (pages 2 – 3)
3. Introduction changed to Background (page 4)
4. Full name of the ethics committee provided (page 14)
5. Conclusion optimized (page 22)
6. List of abbreviations provided (page 22)
7. Authors’ contributions provided (page 23)
8. Competing interests provided (page 23)
9. Acknowledgements provided (page 23)
10. All tables and figures included in the main manuscript (pages 28 – 51)
11. Figures properly formatted according to IforA (see your email 20.02.2015)